Author's response to reviews

Title: Relationship between home care service use and changes in the care needs level of elderly Japanese

Authors:

Gohei Kato (goheik@saitama-med.ac.jp)
Nanako Tamiya (ntamiya@md.tsukuba.ac.jp)
Masayo Kashiwagi (masayo-tky@umin.ac.jp)
Mikiya Sato (mikiya@sato)
Hideto Takahashi (takahasi@md.tsukuba.ac.jp)

Version: 4 Date: 1 December 2009

Author's response to reviews: see over
Gohei KATO
Department of Health Services Research, Graduate School of Comprehensive Human Sciences, University of Tsukuba,
Ibaraki, Japan
Email: goheik@saitama-med.ac.jp
Tel: 81 - (0)49-295-1001
FAX: 81 - (0)49-295-5104

December 1st, 2009
BioMed Central
Scientific Editor
Dr. Hans Zauner

MS: 1010394794218229
Research article
Effect of home care services on the care needs levels of elderly Japanese:
A longitudinal study by Gohei Kato, Nanako Tamiya, Masayo Kashiwagi, Mikiya Sato and Hideto Takahashi
Journal: BMC Geriatrics

Dear Dr. Hans Zauner:
I appreciate your and the reviewers’ helpful advice about my manuscript.

The changes we made were follows.

1. We re-made the table with using table tool on the Word.
2. We removed line numbers.
3. We removed underlining.
4. We renamed ‘Introduction’ to ‘Background’.
5. We removed ‘Additional files’.

I do appreciate for all of your help.

Sincerely yours,

Gohei KATO